Author’s response to reviews

Title: Doctors' opinion on the contribution of coordination mechanisms to improving clinical coordination between primary and outpatient secondary care in the Catalan national health system

Authors:
Marta Beatriz Aller (maller@consorci.org;nmartinez@consorci.org)
Ingrid Vargas (ivargas@consorci.org)
Jordi Coderch (jcoderch@ssibe.cat)
M. Luisa Vázquez (mlvazques@consorci.org)

Version: 2 Date: 05 Oct 2017

Author’s response to reviews:

We are glad to send you a new version of the manuscript “Doctors' opinion on the contribution of coordination mechanisms to improving clinical coordination between primary and outpatient secondary care in the Catalan national health system”, which contains the changes suggested by editors.